HUANG, L., LIU, Z., LI, H., WANG, Y., LI, Y., ZHU, Y., OOI, M.C.G., AN, J., SHANG, Y., ZHANG, D., CHAN, A. and LI, L. 2020. The silver lining of COVID-19: estimation of short-term health impacts due to lockdown in the Yangtze River Delta region, China. *GeoHealth* [online], 4(9), e2020gh000272. Available from: <u>https://doi.org/10.1029/2020GH000272</u>

The silver lining of COVID-19: estimation of shortterm health impacts due to lockdown in the Yangtze River Delta region, China.

HUANG, L., LIU, Z., LI, H., WANG, Y., LI, Y., ZHU, Y., OOI, M.C.G., AN, J., SHANG, Y., ZHANG, D., CHAN, A. and LI, L.

2020

© 2020. The Authors.



This document was downloaded from https://openair.rgu.ac.uk





GeoHealth

RESEARCH ARTICLE

10.1029/2020GH000272

Special Section:

The COVID-19 Pandemic: Linking Health, Society and Environment

Key Points:

- PM_{2.5} concentration decreased substantially during COVID-19 lockdown in the Yangtze River Delta Region, China
- Number of PM_{2.5}-related premature death decreased due to lockdown
- Avoided premature mortality is mostly contributed by reduced death associated with stroke, ischemic heart disease, and chronic obstructive pulmonary disease

Supporting Information:

Supporting Information S1

Correspondence to:

L. Li, lily@shu.edu.cn

Citation:

Huang, L., Liu, Z., Li, H., Wang, Y., Li, Y., Zhu, Y., et al. (2020). The silver lining of COVID-19: Estimation of short-term health impacts due to lockdown in the Yangtze River Delta region, China. *GeoHealth*, 4, e2020GH000272. https://doi.org/ 10.1029/2020GH000272

Received 17 MAY 2020 Accepted 30 JUN 2020 Accepted article online 7 JUL 2020

Author Contributions:

Conceptualization: Ling Huang, Li Li Data curation: Hongli Li, Yonghui Zhu, Dongping Zhang Formal analysis: Ziyi Liu, Hongli Li Methodology: Ling Huang, Yumin Li, Jing An, Yu Shang, Li Li Visualization: Ziyi Liu Writing - original draft: Ling Huang (continued)

©2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

The Silver Lining of COVID-19: Estimation of Short-Term Health Impacts Due to Lockdown in the Yangtze River Delta Region, China

Ling Huang^{1,2}, Ziyi Liu^{1,2}, Hongli Li^{1,2}, Yangjun Wang^{1,2}, Yumin Li³, Yonghui Zhu^{1,2}, Maggie Chel Gee Ooi^{4,5}, Jing An^{1,2}, Yu Shang^{1,2}, Dongping Zhang^{1,2}, Andy Chan⁴, and Li Li^{1,2}

¹School of Environmental and Chemical Engineering, Shanghai University, Shanghai, China, ²Key Laboratory of Organic Compound Pollution Control Engineering (MOE), Shanghai University, Shanghai, China, ³SILC Business School, Shanghai University, Shanghai, China, ⁴Department of Civil Engineering, University of Nottingham Malaysia, Semenyih, Selangor, Malaysia, ⁵Institute of Climate Change (IPI), National University of Malaysia (UKM), Bangi, Selangor, Malaysia

Abstract The outbreak of COVID-19 in China has led to massive lockdowns in order to reduce the spread of the epidemic and control human-to-human transmission. Subsequent reductions in various anthropogenic activities have led to improved air quality during the lockdown. In this study, we apply a widely used exposure-response function to estimate the short-term health impacts associated with PM2.5 changes over the Yangtze River Delta (YRD) region due to COVID-19 lockdown. Concentrations of PM25 during lockdown period reduced by 22.9% to 54.0% compared to pre-lockdown level. Estimated PM_{2 5}related daily premature mortality during lockdown period is 895 (95% confidential interval: 637-1,081), which is 43.3% lower than pre-lockdown period and 46.5% lower compared with averages of 2017–2019. According to our calculation, total number of avoided premature death aassociated with PM2.5 reduction during the lockdown is estimated to be 42.4 thousand over the YRD region, with Shanghai, Wenzhou, Suzhou (Jiangsu province), Nanjing, and Nantong being the top five cities with largest health benefits. Avoided premature mortality is mostly contributed by reduced death associated with stroke (16.9 thousand, accounting for 40.0%), ischemic heart disease (14.0 thousand, 33.2%), and chronic obstructive pulmonary disease (7.6 thousand, 18.0%). Our calculations do not support or advocate any idea that pandemics produce a positive note to community health. We simply present health benefits from air pollution improvement due to large emission reductions from lowered human and industrial activities. Our results show that continuous efforts to improve air quality are essential to protect public health, especially over city-clusters with dense population.

Plain Language Summary The outbreak of COVID-19 is tragic and has caused tremendous impacts on people's life across the world. In late January 2020, massive lockdown was taken in China in order to prevent the spread of the pandemic and control human-to-human transmissions. As a result of reduced human activities, air quality was shown to improve substantially over China. A natural follow-up question would be what is the short-term health impact associated with the changes of air quality during the lockdown. This study attempted to quantify the number of avoided death due to reduced PM_{2.5} concentrations over the Yangtze River Delta Region in China, a region with dense population and used to suffer from severe air pollution problems in the past.

1. Introduction

The outbreak of the tragic Coronavirus disease 2019 (COVID-19) by the end of 2019 has caused tremendous impacts on people's life around the world. At the time of this writing (6th May 2020), COVID-19 has made more than 3.6 million people sick and led to more than 257,301 deaths worldwide (https://www.statista. com/statistics/, last access on 6th May 2020). During its peak, the pandemic at one point caused over 15,000 new confirmed cases in China in just one single day back in February, and presently in May very few new local infections are reported in China (http://www.nhc.gov.cn/, last access on 6th May 2020). The effective containment of COVID-19 within China is mostly attributed to a series of prevention and control measures implemented rapidly by the Chinese government. Starting from late January 2020, national emergency response policies were launched in China in order to reduce the intensity of the spread of the



Writing – review & editing: Yangjun Wang, Maggie Chel Gee Ooi, Andy Chan, Li Li

epidemic to slow down the increase of number of new cases, including but not limited to: schools shut down, traffic strictly restricted, industries and construction activities suspended, mass gatherings and events canceled or suspended, and social distancing become the new norm. As a result of the massive lockdown, emissions of primary air pollutants from various human and industrial activities decreased substantially and PM_{2.5} concentrations during COVID-19 lockdown in China has been shown to be much better than previous years during the same period (NASA, 2020; P. Wang et al., 2020). It is well known that poor air quality, with $PM_{2.5}$ (particulate matters with aerodynamic diameters less than 2.5 μ m) being a key criteria pollutant, could have adverse health impacts (Boldo et al., 2011; Cao et al., 2012; Song et al., 2017) and lead to premature mortality (Fang et al., 2016; Liu et al., 2016; Lu, Zhou, et al., 2015). An integrated exposure risk (IER) model for PM_{2.5} exposure-response function is widely used to estimate the premature mortality attributed to PM_{2.5} exposure. For example, Maji et al. (2018) estimates PM₂ s-related long-term premature mortality for 161 cities in China for year 2015 as well as the potential health benefits of air pollution control policies for year 2020. Y. Wang et al. (2020) calculate the number of premature death due to acute and chronic exposure of ambient $PM_{2.5}$ in China during 2013–2017. With substantial reductions in $PM_{2.5}$ concentrations due to COVID-19 lockdown, a follow-up question is what are the health impacts of the short-term changes in air quality.

The Yangtze River Delta (YRD) region is one of the most economic developed and populated regions in China. In the past, the YRD region has frequently experienced heavy haze pollution (Cheng et al., 2013; Wang et al., 2015). With various control strategies continuously being carried out, the overall air quality over the YRD region has greatly improved for the past few years (Ministry of Ecology and Environment of China, 2019). According to the latest report released by the Ministry of Ecology and Environment of China (2020), 40 out of 41 cities in the YRD region has successfully met the goals of reducing PM_{2.5} concentrations during the 2019–2020 fall and winter season.

In our most recent study (L. Li et al., 2020), we investigate the air quality changes over the YRD region due to lowered human activities during COVID-19 lockdown using multipollutant observations and photochemical model simulations. In this follow-up study, we attempt to quantify the short-term health impacts associated with $PM_{2.5}$ changes over the YRD region due to COVID-19 lockdown. We estimate the premature mortality associated with $PM_{2.5}$ exposure before lockdown and during lockdown periods. Utilizing simulated results based on an integrated meteorology and air quality modeling system, we estimate the number of avoided premature death due to lowered $PM_{2.5}$ concentrations during COVID-19 lockdown over the YRD region. Methods and results from our previous study (L. Li et al., 2020) are partially adopted in this study to support health related estimation.

2. Materials and Methods

2.1. Quantitative Analysis PM_{2.5} Changes Due to COVID-19 Lockdown

The YRD region, consisting of Shanghai, Jiangsu, Zhejiang, and Anhui province (Figure 1), is one of the most economic developed and populated regions in China. On 23rd January 2020, being one of three earliest provinces (the other two being Hunan and Guangdong provinces), Zhejiang province (located in south of the YRD region) announced provincial lockdown as "Level I" (particularly serious) response, followed by Shanghai and Anhui province on the next day and Jiangsu province 2 days later. Coincided with the Chinese Spring Festival (24th January to 1st February 2020), all kinds of human activities were greatly reduced during Level I response period. With the epidemic gradually controlled, emergency response in Anhui and Jiangsu province was downgraded to Level II (serious) on 25th February, followed by Zhejiang province on 2nd March. Shanghai announced Level II response on 24th March due to high numbers of imported infectious cases. Same as previous study, we define pre-lockdown period as 1st January to 23rd January, Level I response period as 24th January to 25th February, and Level II response period as 26th February to 31st March.

To quantify the changes of air quality caused by reduced human activities during COVID-19 lockdown, the integrated Weather Research Forecasting model (WRF)-Comprehensive Air Quality Model with Extensions (CAMx) modeling system is used (L. Li et al., 2020). Details of model configurations and input data can be found in L. Li et al. (2020) and are briefly summarized here. The integrated WRF/CAMx model is applied to simulate air quality over the YRD region (Figure 1) during pre-lockdown, Level I response, and Level II



Figure 1. Location of the Yangtze River Delta (YRD) region with city-level population.

response periods. Two parallel simulations are conducted with two sets of anthropogenic emissions while keeping all other inputs and model configurations identical. For the base case simulation, the baseline emissions (i.e., emissions from normal activities assuming no lockdown) are used. For the COVID-19 scenario, emissions estimated based on reduced human activities due to lockdown are applied. For emission reductions outside the YRD region during lockdown, we applied the reduction ratio used by P. Wang et al. (2020). The relative improvement factor (RF) is defined as the ratio of simulated concentrations between the two scenarios and is applied to the observed concentrations to obtain the theoretical concentrations of air pollutants that would be if there is no lockdown. Results of model performance evaluation of the COVID-19 scenario show acceptable agreement between simulated and observed results (L. Li et al., 2020).

2.2. Premature Mortality Due to Short-Term PM_{2.5} Exposure

We estimate the premature mortality due to ambient $PM_{2.5}$ exposure based on a widely used log-linear exposure-response function below (Fang et al., 2016; Gao et al., 2016):

$$Y = \sum_{k} P \times \left(1 - e^{-\beta_k (C - C_0)} \right) \times R_k, \tag{1}$$

where Y is the number of premature deaths caused by ambient $PM_{2.5}$ exposure due to five leading causes (k = 5): cerebrovascular disease (stroke), ischemic heart disease (IHD), chronic obstructive pulmonary disease (COPD), lung cancer (LC) for adults (\geq 25 years), and acute lower respiratory infection (ALRI) for infants (<5 years). β is the cause-specific exposure-response coefficients and values reported from a meta-analysis study (Lu, Xu, et al., 2015) are utilized in this study. For an increase of 10 μ g/m³ PM_{2.5}, β is 0.63% [95% confidential interval (CI): 0.35-0.9%] for cardiovascular disease (i.e., stoke, IHD) and 0.75% (95% CI: 0.39-1.11%) for respiratory disease (i.e., COPD, ALRI, LC). The baseline incidence rate (R) at provincial level is obtained from the Sixth National Population Census (http://www.stats.gov.cn/ tisj/pcsj/rkpc/6rp/indexch.htm, last access on 20th April 2020) and the contribution of individual disease to total mortality are based on the national estimates from the Global Burden of Diseases (GBD) project of Institute for Health Metrics and Evaluation (IHME) and Health Effects Institute (HEI) for year 2017 (https://vizhub.healthdata.org/gbd-compare/, last access on 23rd April 2020). According to GBD study, stroke, IHD, COPD, LC, and ALRI contribute 20.2%, 16.7%, 9.2%, 6.4%, and 1.7% of total deaths in China for year 2017. P is the exposed population for each city in the YRD region and is obtained from statistical yearbooks for year 2018. The exposed PM_{2.5} concentration in Equation 1 is the PM_{2.5} concentrations averaged during each period. The threshold $PM_{2.5}$ concentration C_0 per World Health





Figure 2. Relative changes of observed $PM_{2.5}$ concentrations at 202 monitoring sites over the YRD region before (1st January to 23rd January 2020) and during lockdown (24th January to 31st March 2020).

Organization (WHO) air quality guidelines (WHO, 2005) of 25 μ g/m³ is used. We calculate the number of premature mortality using observed PM_{2.5} concentrations during pre-lockdown, Level I, and Level II period of 2017–2020 to show trends over the past 4 years. Adjusted PM_{2.5} concentrations calculated based on the RF method (section 2.1) are used to calculate the number of premature mortality assuming no lockdown occurred. The differences between results obtained based on observed and adjusted PM_{2.5} concentrations illustrate the health impacts due to changes in air quality during lockdown periods.

3. Results and Discussions

3.1. Changes of PM_{2.5} Concentrations During COVID-19 Lockdown

Figure 2 shows the relative changes in observed PM_{2.5} concentrations before lockdown (1st January to 23rd January 2020) and lockdown period (24th January to 31st March 2020) for the YRD region. Averaged PM_{2.5} concentrations exhibit ubiquitous decreases across the YRD region during lockdown period with relative reductions by 22.9% (Huangshan in southern Anhui province) ~ 54.0% (Yangzhou in Jiangsu province). During pre-lockdown period, Huaibei (in northern Anhui province) had the highest PM_{2.5} concentrations of 113.3 μ g/m³ among all cities over the YRD region. The highest city-level PM_{2.5} concentrations dropped to 64.9 μ g/m³ in Bozhou (in northern Anhui province) during lockdown periods. Compared with same periods of previous years (averages of 2017–2019), PM_{2.5} concentrations during Level I response period decreased by 16.4% to 56.3% and by 4.8% to 50.9% during Level II response period (supporting information Tables S1–S4).

Figure 3 compares the spatial distribution of $PM_{2.5}$ concentrations with and without COVID-19 lockdown. If no lockdown occurred (i.e., emissions were at baseline level), $PM_{2.5}$ concentrations are estimated to be 27.1 (Zhoushan) to 97.5 µg/m³ (Bozhou), as compared to 18.7 (Zhoushan) to 64.9 µg/m³ (Bozhou). In Shanghai, Hangzhou, Nanjing, and Hefei, $PM_{2.5}$ concentrations are estimated to be 52.4, 43.1, 53.5, and 70.8 µg/m³, which is 64.8%, 50.6%, 58.4%, and 85.1% higher than lockdown values.



Figure 3. Spatial distribution of PM_{2.5} concentrations during lockdown period (left) and adjusted PM_{2.5} concentrations assuming no lockdown (right) over the YRD region.





Figure 4. Premature mortality due to LC, stroke, IHD, COPD during pre-lockdown, Level I, and Level II periods of 2017–2020 (data for ALRI are not shown due to low numbers). Estimated premature mortality with the assumption of no-lockdown is also shown for Level I and Level II.

3.2. Premature Mortality Attributable to Short-Term PM_{2.5} Exposure

Ambient PM_{2.5} exposure leads to higher mortality in infants (<5 years) from ALRI and in adults (\geq 25 years) due to stroke, IHD, COPD, and LC. We calculate the premature mortality due to above-mentioned causes based on the health impact function (Equation 1) over the YRD region during pre-lockdown, Level I, and Level II period of 2017-2020 (Figure 4). During the pre-lockdown period, the total premature mortality attributed to PM_{2.5} exposure are relatively consistent during 2017–2020 and the number in 2020 is 36.4 thousand (95% CI: 30.4-38.8 thousand) for the whole YRD region. Stroke and IHD contribute to 14.4 thousand (95% CI: 12.0-15.4 thousand) and 11.9 thousand (95% CI: 10.0-12.7 thousand) premature death, together accounting for 72.2% of total PM2.5-related premature death. COPD, LC, and ALRI contribute the rest 27.8% of PM2.5-related death, each causing 6.8 thousand (95% CI: 5.7-7.2 thousand), 3.2 thousand (95% CI: 2.7-3.4 thousand), and 0.05 thousand (95% CI: 0.04-0.06 thousand) premature death during the pre-lockdown period. During Level I and Level II response periods, while the premature morality due to PM_{2.5} exposure during 2017–2019 fluctuated a little bit, a sharp decrease is observed for year 2020. The total PM2.5-related premature mortality during 2020 Level I and Level II period is 33.2 thousand (95% CI: 23.9-38.5 thousand) and 27.7 thousand (95% CI: 18.6-33.8 thousand), which dropped by 32.3% and 47.7% compared to the 2017-2019 average values. The relative contributions from different diseases remain unchanged since the only changing variable here is the $PM_{2.5}$ concentration.

In order to directly compare the premature mortality before and during lockdown (Level I plus Level II period), we present the daily premature mortality in Figure S1. During 2017–2019, the daily premature morality dropped by 3.2% to 12.7% before and during lockdown. For 2020, the daily premature mortality across the YRD region during pre-lockdown is 1.5 thousand (95% CI: 1.3–1.6 thousand) and 0.9 thousand (95% CI: 0.6–1.1 thousand) during lockdown (Level I + Level II), representing a sharp decrease by 43.3%. The significant reduction in premature mortality during lockdown periods, whether it is compared to the pre-lockdown periods of the same year, or the same periods of historical years, indicate substantial health benefits associated with lowered PM_{2.5} concentrations due to COVID-19 lockdown.

Tables S1–S4 show the city-level premature mortality by period and year. With the same base incidence rate (R) being applied for all cities, city-level premature mortality depends on the exposed population and the $PM_{2.5}$ concentration for that city. With the largest population (2418.3 thousand in 2018) of all cities in the YRD region (Figure 5), Shanghai has the highest $PM_{2.5}$ -related premature mortality during 2017–2019, even though its averaged $PM_{2.5}$ concentration is ranked the sixth, ninth, and ninth from the bottom during January to March of 2017, 2018, and 2019 (Figure 5). On the other hand, high premature mortality for cities like Bozhou and Suqian in Anhui province (e.g., each is ranked third and seventh during Level II period in





Figure 5. City-level population, averaged $PM_{2.5}$ concentration, and population weighted $PM_{2.5}$ exposure (calculated as city-level population multiplied with $PM_{2.5}$ concentration divided by averaged population across all cities) during January to March of 2017–2019 for 41 cities over the YRD region.

2020 in terms of premature mortality) is more associated with the high $PM_{2.5}$ concentrations, which ranked third and seventh in terms of average $PM_{2.5}$ concentration and only thirteenth and nineteenth in terms of population. Cities with small population and low $PM_{2.5}$ concentrations, for example, Huangshan (in Anhui province), have the lowest premature mortality. During 2020 pre-lockdown, daily premature mortality due to $PM_{2.5}$ exposure in Shanghai is 145 (95% CI: 112–158). This number drops to 84 (95% CI: 53–106) during Level I period and 25 (95% CI: 14–35) during Level II period, each representing a decrease by 42.1% and 82.8% compared to pre-lockdown level. During 2020 Level II period, the number of premature death due to $PM_{2.5}$ exposure in Shanghai drops to the twelfth place due to a sharp decrease in $PM_{2.5}$ concentrations during Level II (decrease by 24.0%). The number of cities with zero $PM_{2.5}$ -related premature mortality (i.e., $PM_{2.5}$ concentration below threshold value) increases from zero during 2020 pre-lockdown period to five (Huangshan in Anhui province, Lishui, Quzhou, Wenzhou, and Zhoushan in Zhejiang province) during Level I and six (Huzhou, Jiaxing, Lishui, Ningbo, Wenzhou, and Zhoushan in Zhejiang province) during Level II period.

3.3. Avoided Premature Mortality Due to PM_{2.5} Improvement During COVID-19 Lockdown

An integrated WRF/CAMx model is used to estimate $PM_{2.5}$ concentrations that would be during Level I and Level II period if no lockdown occurred. The potential health benefits due to lockdown are estimated as the differences of premature mortality calculated based on the observed $PM_{2.5}$ concentrations and the simulated "no lockdown" concentrations during the same periods. If no lockdown occurred, the total premature death due to $PM_{2.5}$ exposure over the YRD region would be 50.2 thousand (95% CI: 42.0–53.3 thousand) during Level I period and 52.9 thousand (95% CI: 40.6–58.5 thousand) during Level II period. These two numbers are lower than the corresponding values of 2017–2019 (except for Level I in 2017 and 2019). The total avoided premature mortality over the YRD region is 17.1 thousand during Level I and 25.1 thousand during Level II (Table S5), each representing 51.5% and 90.6% of current premature mortality rate due to $PM_{2.5}$ exposure. In terms of diseases, avoided premature morality due to IHD and stroke are 14.0 thousand and 16.9 thousand, each accounting for 33.2% and 40.1% of current base incident mortality rate due to $PM_{2.5}$. Avoided premature mortality due to COPD, LC, and ALRI are 7.6 thousand, 3.6 thousand, and 0.06 thousand.

Figure 6 shows the city-level total avoided premature mortality during Level I and Level II (avoided premature mortality associated with different diseases are shown in Figure S2). The number of avoided premature





Figure 6. Spatial distribution of avoided total premature death during Level I and Level II.

death for different cities depends on the base population and the changes in $PM_{2.5}$ concentrations due to lockdown (Table S5). During Level I period, the top five cities with largest avoided premature death are Shanghai (1,932), Wenzhou (1,124), Suzhou (Jiangsu province; 1,118), Ningbo (934), and Hangzhou (934). During Level II period, the top five cities with largest avoided premature death are Shanghai (3,502), Wenzhou (1,628), Suzhou (Jiangsu province; 1,542), Nantong (1,270), and Nanjing (1,227). It should be noted that the higher number of avoided premature death during Level II period over Level I period is simply because Level II covers more days (36 days) than Level I (31 days). The daily avoided deaths during Level I are higher than Level II due to more restricted transportation and industrial activities.

3.4. Premature Mortality Due to Residual PM_{2.5} During COVID-19 Lockdown

Although the $PM_{2.5}$ concentrations during COVID-19 lockdown seem to have dropped substantially whether compared with pre-lockdown periods or the same periods of previous years, there are still 36 (during Level I) and 35 (during Level II) out of 41 cities with residual $PM_{2.5}$ concentrations that are higher than the recommended 24-hr $PM_{2.5}$ concentrations of 25 µg/m³ by WHO. Premature mortality caused by $PM_{2.5}$ -exposure during Level I and Level II period is estimated to be 33.2 thousand (95% CI: 23.9–38.4 thousand) and 27.7 thousand (95% CI: 18.6–33.8 thousand), respectively, each representing 31.2% and 22.4% of total death rate over the YRD region. The top five cities with highest premature mortality during Level I are Shanghai (2,687, 95% CI: 1,707–3,385), Fuyang (2,581, 95% CI: 2,197–2,678), Xuzhou (2,229, 95% CI: 1,835–2,348), Suzhou (Jiangsu province, 1,606, 95% CI: 1,021–2,023), and Bozhou (1,599, 95% CI: 1,406–1,639). During Level II period, the top five cities with highest premature mortality are Fuyang (2,647, 95% CI: 1,996–2,929), Xuzhou (2,226, 95% CI: 1,627–2,514), Bozhou (1,587, 95% CI: 1,185–1767), Suzhou (Jiangsu province, 1,459, 95% CI: 1,627–2,514), Bozhou (Anhui province, 1,449, 95% CI: 1,027–1,672).

3.5. Uncertainties

A few of uncertainties exist with our estimation of health impacts, which are also recognized in a couple of similar studies (Liu et al., 2016; Maji et al., 2018; Y. Wang et al., 2020). First and foremost, there exists uncertainties with the parameters (e.g., concentration-response coefficient, threshold concentration) used in the integrated exposure-response function. Although limitation exists, we use values reported by Lu, Xu, et al. (2015), which is developed based on meta-analysis of 59 studies covering 22 cities in Mainland China (three cities in the YRD region), Hong Kong, and Taiwan to reduce uncertainties with the concentration-response coefficient does not vary by age and a provincial-level base incidence rate is used for all cities within the province. Second, we only calculate the premature mortality associated with $PM_{2.5}$ exposure while realizing that synergistic effects exist when exposed to other or multiple pollutants (Apte et al., 2015; Billionnet et al., 2012) and our estimation of premature mortality perhaps represents an underestimation. On top of that, the influences of $PM_{2.5}$ chemical composition, size distribution, and



sources on health impacts (Ostro et al., 2015) are ignored in this study. When we calculate city-level population exposure, we ignore the heterogeneities of ambient $PM_{2.5}$ concentrations and population within the city. We simply use the observed $PM_{2.5}$ concentrations averaged over all monitoring sites within the city as the exposed concentrations. This introduces underestimation when more monitoring sites are located in the rural and less populated areas and overestimation when monitoring sites are more likely to be located in urban and populous areas for a city. Ways to improve the spatial accuracy of health impact estimation include utilizing population distribution with high spatial resolution and interpolated $PM_{2.5}$ concentrations based on networks of low-cost sensors (Cavaliere et al., 2018; J. Li et al., 2020; Holstius et al., 2014) or satellite based data (e.g., the aerosol optical depth, Chen et al., 2019; J. Li et al., 2020). Finally, estimations of avoided premature death assuming no-lockdown are associated with uncertainties with the meteorology and air quality model. This may all be explored in the near-future when more data are available.

4. Conclusions

In this study, we attempt to quantify the health impacts associated with $PM_{2.5}$ improvement due to reduced human and industrial activities during COVID-19 lockdown in the YRD region, a region with heavy air pollution during the past years. As a result of reduced human activities, concentrations of $PM_{2.5}$ during lockdown periods reduced by 22.9% to 54.0% over the YRD region compared to pre-lockdown level. Avoided premature death due to lowered $PM_{2.5}$ concentrations are estimated to be 42.2 thousand over the YRD region, representing 69.3% of the total present premature mortality due to $PM_{2.5}$ exposure. The avoided premature mortality is mostly contributed by reduced death associated with stroke (16.9 thousand, accounting for 40.0%), ischemic heart disease (14.0 thousand, 33.2%), and chronic obstructive pulmonary disease (7.6 thousand, 18.0%). Top five cities with highest avoided premature mortality are Shanghai (5,433 persons), Wenzhou (2,751), Suzhou (in Jiangsu province, 2,660), Nanjing (2,000), and Nantong (1,969).

The outbreak of COVID-19 is disastrous. This study is by no means to suggest that pandemics are bringing a positive effect for health. The passive emission reductions during COVID-19 lockdown provide a good opportunity to show the health impacts related to reduction in $PM_{2.5}$. It merely reinforces our knowledge that $PM_{2.5}$ has detrimental health effects. Results from our study suggested that substantial health benefits could be achieved with reduced $PM_{2.5}$ concentrations by emission reductions, which confirms the importance of recent efforts to mitigate the haze pollution nationwide. However, although $PM_{2.5}$ concentration decreased substantially during lockdown period, the residual $PM_{2.5}$ concentrations are still much higher than the recommended 24-hr standard by WHO. Continuous efforts are needed to reduce emissions in the long term and through the most cost-effective ways in order to protect public health.

Conflict of Interest

The authors declare no conflict of interest relevant to this study.

References

Apte, J. S., Marshall, J. D., Cohen, A. J., & Brauer, M. (2015). Addressing Global Mortality from Ambient PM2.5. Environmental Science & Technology, 49(13), 8057–8066. https://doi.org/10.1021/acs.est.5b01236

- Billionnet, C., Sherrill, D., & Annesi-Maesano, I. (2012). Estimating the health effects of exposure to multi-pollutant mixture. Annals of Epidemiology, 22(2), 126–141. https://doi.org/10.1016/j.annepidem.2011.11.004
- Boldo, E., Linares, C., Lumbreras, J., Borge, R., Narros, A., & Garcia-Perez, J. (2011). Health impact assessment of a reduction in ambient PM2.5 levels in Spain. *Environment International*, 37(2), 342–348. https://doi.org/10.1016/j.envint.2010.10.004

Cao, J., Xu, H., Xu, Q., Chen, B., & Kan, H. (2012). Fine particulate matter constituents and cardiopulmonary mortality in a heavily polluted Chinese city. Environmental Health Perspectives, 120(3), 373–378. https://doi.org/10.1289/ehp.1103671

- Cavaliere, A., Carotenuto, F., Di Gennaro, F., Gioli, B., Gualtieri, G., Martelli, F., et al. (2018). Development of low-cost air quality stations for next generation monitoring networks: Calibration and validation of PM2.5 and PM10 Sensors. Sensors (Basel), 18(9), 2843. https:// doi.org/10.3390/s18092843
- Chen, Z.-Y., Zhang, T.-H., Zhang, R., Zhu, Z.-M., Yang, J., Chen, P.-Y., et al. (2019). Extreme gradient boosting model to estimate PM2.5 concentrations with missing-filled satellite data in China. Atmospheric Environment, 202, 180–189. https://doi.org/10.1016/j. atmosenv.2019.01.027
- Cheng, Z., Wang, S., Jiang, J., Fu, Q., Chen, C., Xu, B., et al. (2013). Long-term trend of haze pollution and impact of particulate matter in the Yangtze River Delta, China. *Environmental Pollution*, *182*, 101–110. https://doi.org/10.1016/j.envpol.2013.06.043

Fang, D., Wang, Q., Li, H., Yu, Y., Lu, Y., & Qian, X. (2016). Mortality effects assessment of ambient PM2.5 pollution in the 74 leading cities of China. Science of the Total Environment, 569-570, 1545–1552. https://doi.org/10.1016/j.scitotenv.2016.06.248

Acknowledgments

This study is financially sponsored by the Shanghai Science and Technology Innovation Plan (19DZ1205007), National Natural Science Foundation of China (41875161), Shanghai International Science and Technology Cooperation Fund (19230742500), National Key R&D Program of China (2018YFC0213600), and Shanghai Sail Program (19YF141560). The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper. Air quality data from http:// 106.37.208.233:20035 (China Environmental Monitoring Service). The baseline incidence rate (R) at provincial level is obtained from http:// www.stats.gov.cn/tjsj/pcsj/rkpc/6rp/ indexch.htm (The Sixth National Population Census). The contribution of individual disease to total mortality are based on the national estimates from https://vizhub.healthdata.org/ gbd-compare/ (The Global Burden of Diseases [GBD] project of Institute for Health Metrics and Evaluation [IHME] and Health Effects Institute [HEI] for year 2017).



- Gao, J., Yuan, Z., Liu, X., Xia, X., Huang, X., & Dong, Z. (2016). Improving air pollution control policy in China—A perspective based on cost-benefit analysis. *Science of the Total Environment*, 543(Pt A), 307–314. https://doi.org/10.1016/j.scitotenv.2015.11.037
- Holstius, D. M., Pillarisetti, A., Smith, K. R., & Seto, E. (2014). Field calibrations of a low-cost aerosol sensor at a regulatory monitoring site in California. Atmospheric Measurement Techniques Discussions, 7(1), 605–632. https://doi.org/10.5194/amtd-7-605-2014
- Li, J., Zhang, H., Chao, C.-Y., Chien, C.-H., Wu, C.-Y., Luo, C. H., et al. (2020). Integrating low-cost air quality sensor networks with fixed and satellite monitoring systems to study ground-level PM2.5. Atmospheric Environment, 223, 117293. https://doi.org/10.1016/j. atmosenv.2020.117293
- Li, L., Li, Q., Huang, L., Wang, Q., Zhu, A., Xu, J., et al. (2020). Air quality changes during the COVID-19 lockdown over the Yangtze River Delta region: An insight into the impact of human activity pattern changes on air pollution variations. *Science of the Total Environment*, 732, 139282. https://doi.org/10.1016/j.scitotenv.2020.139282
- Liu, J., Han, Y., Tang, X., Zhu, J., & Zhu, T. (2016). Estimating adult mortality attributable to PM2.5 exposure in China with assimilated PM2.5 concentrations based on a ground monitoring network. *Science of the Total Environment*, 568, 1253–1262. https://doi.org/ 10.1016/j.scitotenv.2016.05.165
- Lu, F., Xu, D., Cheng, Y., Dong, S., Guo, C., Jiang, X., & Zheng, X. (2015). Systematic review and meta-analysis of the adverse health effects of ambient PM2.5 and PM10 pollution in the Chinese population. *Environmental Research*, 136, 196–204. https://doi.org/10.1016/j. envres.2014.06.029

Lu, F., Zhou, L., Xu, Y., Zheng, T., Guo, Y., Wellenius, G. A., et al. (2015). Short-term effects of air pollution on daily mortality and years of life lost in Nanjing, China. Science of the Total Environment, 536, 123–129. https://doi.org/10.1016/j.scitotenv.2015.07.048

Maji, K. J., Dikshit, A. K., Arora, M., & Deshpande, A. (2018). Estimating premature mortality attributable to PM2.5 exposure and benefit of air pollution control policies in China for 2020. Science of the Total Environment, 612, 683–693. https://doi.org/10.1016/j. scitotenv.2017.08.254

Ministry of Ecology and Environment of China. (2019). China Air Quality Improvement Report (2013–2018). http://www.mee.gov.cn/ xxgk2018/xxgk/xxgk15/201906/t20190606_705778.html, last access on May 6th, 2020

- Ministry of Ecology and Environment of China. (2020). Reports on the Completion of Environmental Air Quality Targets for the Autumn and Winter of 2019–2020 in Key Areas, http://www.mee.gov.cn/xxgk2018/xxgk/xxgk15/202004/t20200422_775917.html, last access on May 6th, 2020
- Ostro, B., Hu, J., Goldberg, D., Reynolds, P., Hertz, A., Bernstein, L., & Kleeman, M. J. (2015). Associations of mortality with long-term exposures to fine and ultrafine particles, species and sources: Results from the California teachers study cohort. *Environmental Health Perspectives*, 123(6), 549–556. https://doi.org/10.1289/ehp.1408565
- Song, C., He, J., Wu, L., Jin, T., Chen, X., Li, R., et al. (2017). Health burden attributable to ambient PM_{2.5} in China. *Environmental Pollution*, 223, 575–586. https://doi.org/10.1016/j.envpol.2017.01.060
- Wang, M., Cao, C., Li, G., & Singh, R. P. (2015). Analysis of a severe prolonged regional haze episode in the Yangtze River Delta, China. *Atmospheric Environment*, 102, 112–121. https://doi.org/10.1016/j.atmosenv.2014.11.038
- Wang, P., Chen, K., Zhu, S., Wang, P., & Zhang, H. (2020). Severe air pollution events not avoided by reduced anthropogenic activities during COVID-19 outbreak. Resources. *Conservation and Recycling*, 158, 104814. https://doi.org/10.1016/j.resconrec.2020.104814
- Wang, Y., Wild, O., Chen, H., Gao, M., Wu, Q., Qi, Y., et al. (2020). Acute and chronic health impacts of PM2.5 in China and the influence of interannual meteorological variability. Atmospheric Environment, 229, 117397. https://doi.org/10.1016/j.atmosenv.2020.117397